Author's response to reviews

Title: Impact of socioeconomic deprivation on cause of death in Severe Mental Illness

Authors:

  Julie Langan Martin (julie.langan@glasgow.ac.uk)
  Gary McLean (gary.mclean@glasgow.ac.uk)
  John Park (john.park@ggc.scot.nhs.uk)
  Daniel J Martin (daniel.martin@glasgow.ac.uk)
  Moira Connolly (moira.connolly@ggc.scot.nhs.uk)
  Stewart W Mercer (stewart.mercer@glasgow.ac.uk)
  Daniel J Smith (daniel.smith@glasgow.ac.uk)

Version: 2  Date: 1 August 2014

Author's response to reviews:

Attached are our detailed responses to the reviewers.

Many thanks

Julie Langan Martin